Author's response to reviews

Title: Effects of Abciximab on Key Pattern of Human Coronary Restenosis in Vitro: Impact of the SI/MPL-Ratio

Authors:

  Rainer Voisard (rainer.voisard@uni-ulm.de)
  Mustafa Alan (mustafa.alan@uni-ulm.de)
  Lutz von Mueller (lutz.vonmueller@uni-ulm.de)
  Regine Baur (regine.baur@uni-ulm.de)
  Vinzenz Hombach (vinzenz.hombach@uni-ulm.de)

Version: 4  Date: 20 March 2006

Author's response to reviews:

Ulm, Germany, march 20, 2006

Dear BMC Editorial Team,

enclosed please find the accepted manuscript entitled: "Effects of abciximab on key pattern of human coronary restenosis in vitro: impact of the SI/MPL-ratio."

The final queries of the editorial team have been carefully addressed.

With best wishes

Rainer Voisard, MD
Resident Cardiologist